Author's response to reviews

Title: One year follow-up of a pragmatic multi-centre randomised controlled trial of a group-based fatigue management programme (FACETS) for people with Multiple Sclerosis.

Authors:

Peter W Thomas (pthomas@bournemouth.ac.uk)
Sarah Thomas (saraht@bournemouth.ac.uk)
Paula Kersten (paula.kersten@aut.ac.nz)
Rosemary Jones (rosemary.jones@nbt.nhs.uk)
Vicky Slingsby (vicky.slingsby@poole.nhs.uk)
Alison Nock (alison.nock@poole.nhs.uk)
Angela Davies-Smith (angela.davies-smith@nbt.nhs.uk)
Roger Baker (rbaker@bournemouth.ac.uk)
Kathleen T Galvin (K.T.Galvin@hull.ac.uk)
Charles Hillier (charles.hillier@poole.nhs.uk)

Version: 4 Date: 1 May 2014

Author's response to reviews: see over
Dr Calabrese

Thank you for the provisional acceptance for our paper subject to an editorial amendment. Please find below the editorial comment with our response. Please do not hesitate to contact us should you require any further clarification.

Yours sincerely

Sarah Thomas
Senior Research Fellow

**Editor’s comments**
We would be grateful if you could please reiterate in your methods that written informed consent was obtained from the participants and include the full names of the ethics review boards that approved the study.

*We have added this information as a tracked changed and re-uploaded the manuscript.*